Author's response to reviews

Title: Delays, interruptions, and losses from prevention of mother-to-child transmission of HIV services during antenatal care in Johannesburg, South Africa: A cohort analysis

Authors:

Kathryn Schnippel (kschnipp@gmail.com)
Constance Mongwenyana (cmongwenyana@heroza.org)
Lawrence Long (llong@heroza.org)
Bruce Larson (blarson@bu.edu)

Version: 4 Date: 16 December 2014

Author's response to reviews:

December 2014
Miss Sheryl Ramos
editorial@biomedcentral.com

RE: Submission of revised manuscript MS: 2123159240137973

Dear Miss Ramos:

On behalf of my co-authors I am pleased to submit the revisions to the manuscript titled: Delays, interruptions, and losses from prevention of mother-to-child transmission of HIV services during antenatal care in Johannesburg, South Africa: A cohort analysis as per the comments received from the two reviewers.

The revised manuscript and the manuscript with the revisions indicated using track changes in MS Word have both been uploaded to the editorial manager, as per the instructions received in this regard. In the table below, we respond to the comments received from the peer reviewers and also indicate where the text was changed according to the suggestions.

We would thank the journal and the reviewers for the useful comments and suggestions on this manuscript. We believe that these suggestions and associated revisions and additions have strengthened the analysis.

Regards,

Kathryn Schnippel, kschnipp@gmail.com mailto:kschnipp@heroza.org
Health Economics and Epidemiology Research Office, Department of Internal Medicine, Faculty of Health Sciences, University of the Witwatersrand, Johannesburg, South Africa